\documentclass[kind=GCE,a4j,bachelor,landscape,viewer,color,check]{edb-syl}
\def\�N�x{2011}
\def\�a�N�x{23}
\begin{document}

\Englishfalse
%%%% 0:220853
\begin{edb:lecture}%
\EdbEID{220853}\EdbEOID{561667}%
\���false%
\����ID{218270:220853}
\���ƊT�v���{166225}%
\����{�p��}{English}%
\���{��Չp��}{Basic English}%
\�Ώ�{�H}{(��B)1�N}%
\�J�u{�O}{��}{5�E6}%
\�S������{\������������{}{���΍u�t}{\EdbSMG{�j���`���b�N}{}{�A�[����}}}%
\EdbColDef{edb:lecture.goal}{1}{E}{Students should come to class and try to use only English, without fear of making mistakes.\ Mistakes will happen, and students should not worry when they make them.\ Enjoy the class, have fun, and use English.}%
\EdbColDef{edb:lecture.goal}{1}{J}{Students should come to class and try to use only English, without fear of making mistakes.\ Mistakes will happen, and students should not worry when they make them.\ Enjoy the class, have fun, and use English.}%
\EdbColDef{edb:lecture.plan}{1}{E}{Introductions}%
\EdbColDef{edb:lecture.plan}{1}{J}{Introductions}%
\EdbColDef{edb:lecture.plan}{2}{E}{Classroom English}%
\EdbColDef{edb:lecture.plan}{2}{J}{Classroom English}%
\EdbColDef{edb:lecture.plan}{3}{E}{What did you do last weekend?}%
\EdbColDef{edb:lecture.plan}{3}{J}{What did you do last weekend?}%
\EdbColDef{edb:lecture.plan}{4}{E}{Which is bigger?}%
\EdbColDef{edb:lecture.plan}{4}{J}{Which is bigger?}%
\EdbColDef{edb:lecture.plan}{5}{E}{Time}%
\EdbColDef{edb:lecture.plan}{5}{J}{Time}%
\EdbColDef{edb:lecture.plan}{6}{E}{I like cheese pizza.\ Me too!}%
\EdbColDef{edb:lecture.plan}{6}{J}{I like cheese pizza.\ Me too!}%
\EdbColDef{edb:lecture.plan}{7}{E}{First, cut some tomatoes.}%
\EdbColDef{edb:lecture.plan}{7}{J}{First, cut some tomatoes.}%
\EdbColDef{edb:lecture.plan}{8}{E}{Mid-Term Test}%
\EdbColDef{edb:lecture.plan}{8}{J}{Mid-Term Test}%
\EdbColDef{edb:lecture.plan}{9}{E}{Future plans}%
\EdbColDef{edb:lecture.plan}{9}{J}{Future plans}%
\EdbColDef{edb:lecture.plan}{10}{E}{How often do you exercise?}%
\EdbColDef{edb:lecture.plan}{10}{J}{How often do you exercise?}%
\EdbColDef{edb:lecture.plan}{11}{E}{Can you swim?}%
\EdbColDef{edb:lecture.plan}{11}{J}{Can you swim?}%
\EdbColDef{edb:lecture.plan}{12}{E}{Fast food}%
\EdbColDef{edb:lecture.plan}{12}{J}{Fast food}%
\EdbColDef{edb:lecture.plan}{13}{E}{Holidays and celebrations}%
\EdbColDef{edb:lecture.plan}{13}{J}{Holidays and celebrations}%
\EdbColDef{edb:lecture.plan}{14}{E}{I Prefer dark chocolate}%
\EdbColDef{edb:lecture.plan}{14}{J}{I Prefer dark chocolate}%
\EdbColDef{edb:lecture.plan}{15}{E}{Can you speak any other languages?}%
\EdbColDef{edb:lecture.plan}{15}{J}{Can you speak any other languages?}%
\EdbColDef{edb:lecture.plan}{16}{E}{Final Test}%
\EdbColDef{edb:lecture.plan}{16}{J}{Final Test}%
\EdbColDef{edb:lecture.style}{1}{@}{101765}%
\EdbColDef{edb:lecture.style}{1}{E}{Speaking���S}%
\EdbColDef{edb:lecture.style}{1}{J}{Speaking���S}%
\EdbColDef{edb:lecture.unit}{1}{E}{1}%
\EdbColDef{edb:lecture.unit}{1}{J}{1}%
\EdbColDef{edb:lecture.target}{1}{E}{Students will have a chance to practice using English.}%
\EdbColDef{edb:lecture.target}{1}{J}{Students will have a chance to practice using English.}%
\EdbColDef{edb:lecture.outline}{1}{E}{Students will discuss a different topic each week, practicing the English necessary for each week's lesson.}%
\EdbColDef{edb:lecture.outline}{1}{J}{Students will discuss a different topic each week, practicing the English necessary for each week's lesson.}%
\EdbColDef{edb:lecture.keyword}{1}{E}{Practical English; conversation}%
\EdbColDef{edb:lecture.keyword}{1}{J}{Practical English; conversation}%
\EdbColDef{edb:lecture.notice}{1}{E}{Have fun,{\EdbZSp} enjoy and speak English!}%
\EdbColDef{edb:lecture.notice}{1}{J}{Have fun,{\EdbZSp} enjoy and speak English!}%
\EdbColDef{edb:lecture.evaluation}{1}{E}{Students' grades will be as follows:{\EdbZSp} Attendance = 50\% of students' grades.\ Mid- term test = 25\% of students' grades.\ Final test = 25\% of students' grades.\ Final grades will consist of an attendance score and test scores.\ (ie.\ 50\% + 25\% + 25\% = 100\%)}%
\EdbColDef{edb:lecture.evaluation}{1}{J}{Students' grades will be as follows:{\EdbZSp} Attendance = 50\% of students' grades.\ Mid- term test = 25\% of students' grades.\ Final test = 25\% of students' grades.\ Final grades will consist of an attendance score and test scores.\ (ie.\ 50\% + 25\% + 25\% = 100\%)}%
\EdbColDef{edb:lecture.evaluation2}{1}{E}{\EdbAlert{��}}%
\EdbColDef{edb:lecture.evaluation2}{1}{J}{��}%
\EdbColDef{edb:lecture.textbook}{1}{E}{\EdbAlert{��}}%
\EdbColDef{edb:lecture.textbook}{1}{J}{��}%
\EdbColDef{edb:lecture.reference}{1}{E}{Practical English Textbook - Basic (\ooalign{\hfill{Y}\hfill\crcr{=}}{}500) - available from teacher,{\EdbZSp} 2nd week}%
\EdbColDef{edb:lecture.reference}{1}{J}{Practical English Textbook - Basic (\ooalign{\hfill{Y}\hfill\crcr{=}}{}500) - available from teacher,{\EdbZSp} 2nd week}%
\EdbColDef{edb:lecture.contact}{1}{J}{\�A����l{}{�j���`���b�N . }{c{\EdbZSp}a{\EdbZSp}n{\EdbZSp}a{\EdbZSp}d{\EdbZSp}a{\EdbZSp}c{\EdbZSp}l{\EdbZSp}u{\EdbZSp}b{\EdbZSp}j{\EdbZSp}p{\EdbZSp}@{\EdbZSp}b{\EdbZSp}l{\EdbZSp}o{\EdbZSp}o{\EdbZSp}m{\EdbZSp}.{\EdbZSp}o{\EdbZSp}c{\EdbZSp}n{\EdbZSp}.{\EdbZSp}n{\EdbZSp}e{\EdbZSp}.{\EdbZSp}j{\EdbZSp}p{\EdbZSp}}}%
\��������{�p��(1)}%
\������{��Չp��}%
\end{edb:lecture}%

\Englishtrue
%%%% 0:220853
\begin{edb:lecture}%
\EdbEID{220853}\EdbEOID{561667}%
\���false%
\����ID{218270:220853}
\���ƊT�v���{166225}%
\����{English}{�p��}%
\���{Basic English}{��Չp��}%
\�Ώ�{�H}{(��B)1�N}%
\�J�u{�O}{��}{5�E6}%
\�S������{\������������{}{Part-time Lecturer}{\EdbGMS{ARLEN}{}{NIMCHUK}}}%
\EdbColDef{edb:lecture.goal}{1}{J}{Students should come to class and try to use only English, without fear of making mistakes.\ Mistakes will happen, and students should not worry when they make them.\ Enjoy the class, have fun, and use English.}%
\EdbColDef{edb:lecture.goal}{1}{E}{Students should come to class and try to use only English, without fear of making mistakes.\ Mistakes will happen, and students should not worry when they make them.\ Enjoy the class, have fun, and use English.}%
\EdbColDef{edb:lecture.plan}{1}{J}{Introductions}%
\EdbColDef{edb:lecture.plan}{1}{E}{Introductions}%
\EdbColDef{edb:lecture.plan}{2}{J}{Classroom English}%
\EdbColDef{edb:lecture.plan}{2}{E}{Classroom English}%
\EdbColDef{edb:lecture.plan}{3}{J}{What did you do last weekend?}%
\EdbColDef{edb:lecture.plan}{3}{E}{What did you do last weekend?}%
\EdbColDef{edb:lecture.plan}{4}{J}{Which is bigger?}%
\EdbColDef{edb:lecture.plan}{4}{E}{Which is bigger?}%
\EdbColDef{edb:lecture.plan}{5}{J}{Time}%
\EdbColDef{edb:lecture.plan}{5}{E}{Time}%
\EdbColDef{edb:lecture.plan}{6}{J}{I like cheese pizza.\ Me too!}%
\EdbColDef{edb:lecture.plan}{6}{E}{I like cheese pizza.\ Me too!}%
\EdbColDef{edb:lecture.plan}{7}{J}{First, cut some tomatoes.}%
\EdbColDef{edb:lecture.plan}{7}{E}{First, cut some tomatoes.}%
\EdbColDef{edb:lecture.plan}{8}{J}{Mid-Term Test}%
\EdbColDef{edb:lecture.plan}{8}{E}{Mid-Term Test}%
\EdbColDef{edb:lecture.plan}{9}{J}{Future plans}%
\EdbColDef{edb:lecture.plan}{9}{E}{Future plans}%
\EdbColDef{edb:lecture.plan}{10}{J}{How often do you exercise?}%
\EdbColDef{edb:lecture.plan}{10}{E}{How often do you exercise?}%
\EdbColDef{edb:lecture.plan}{11}{J}{Can you swim?}%
\EdbColDef{edb:lecture.plan}{11}{E}{Can you swim?}%
\EdbColDef{edb:lecture.plan}{12}{J}{Fast food}%
\EdbColDef{edb:lecture.plan}{12}{E}{Fast food}%
\EdbColDef{edb:lecture.plan}{13}{J}{Holidays and celebrations}%
\EdbColDef{edb:lecture.plan}{13}{E}{Holidays and celebrations}%
\EdbColDef{edb:lecture.plan}{14}{J}{I Prefer dark chocolate}%
\EdbColDef{edb:lecture.plan}{14}{E}{I Prefer dark chocolate}%
\EdbColDef{edb:lecture.plan}{15}{J}{Can you speak any other languages?}%
\EdbColDef{edb:lecture.plan}{15}{E}{Can you speak any other languages?}%
\EdbColDef{edb:lecture.plan}{16}{J}{Final Test}%
\EdbColDef{edb:lecture.plan}{16}{E}{Final Test}%
\EdbColDef{edb:lecture.style}{1}{@}{101765}%
\EdbColDef{edb:lecture.style}{1}{J}{Speaking���S}%
\EdbColDef{edb:lecture.style}{1}{E}{Speaking���S}%
\EdbColDef{edb:lecture.unit}{1}{J}{1}%
\EdbColDef{edb:lecture.unit}{1}{E}{1}%
\EdbColDef{edb:lecture.target}{1}{J}{Students will have a chance to practice using English.}%
\EdbColDef{edb:lecture.target}{1}{E}{Students will have a chance to practice using English.}%
\EdbColDef{edb:lecture.outline}{1}{J}{Students will discuss a different topic each week, practicing the English necessary for each week's lesson.}%
\EdbColDef{edb:lecture.outline}{1}{E}{Students will discuss a different topic each week, practicing the English necessary for each week's lesson.}%
\EdbColDef{edb:lecture.keyword}{1}{J}{Practical English; conversation}%
\EdbColDef{edb:lecture.keyword}{1}{E}{Practical English; conversation}%
\EdbColDef{edb:lecture.notice}{1}{J}{Have fun,{\EdbZSp} enjoy and speak English!}%
\EdbColDef{edb:lecture.notice}{1}{E}{Have fun,{\EdbZSp} enjoy and speak English!}%
\EdbColDef{edb:lecture.evaluation}{1}{J}{Students' grades will be as follows:{\EdbZSp} Attendance = 50\% of students' grades.\ Mid- term test = 25\% of students' grades.\ Final test = 25\% of students' grades.\ Final grades will consist of an attendance score and test scores.\ (ie.\ 50\% + 25\% + 25\% = 100\%)}%
\EdbColDef{edb:lecture.evaluation}{1}{E}{Students' grades will be as follows:{\EdbZSp} Attendance = 50\% of students' grades.\ Mid- term test = 25\% of students' grades.\ Final test = 25\% of students' grades.\ Final grades will consist of an attendance score and test scores.\ (ie.\ 50\% + 25\% + 25\% = 100\%)}%
\EdbColDef{edb:lecture.evaluation2}{1}{J}{��}%
\EdbColDef{edb:lecture.evaluation2}{1}{E}{\EdbAlert{��}}%
\EdbColDef{edb:lecture.textbook}{1}{J}{��}%
\EdbColDef{edb:lecture.textbook}{1}{E}{\EdbAlert{��}}%
\EdbColDef{edb:lecture.reference}{1}{J}{Practical English Textbook - Basic (\ooalign{\hfill{Y}\hfill\crcr{=}}{}500) - available from teacher,{\EdbZSp} 2nd week}%
\EdbColDef{edb:lecture.reference}{1}{E}{Practical English Textbook - Basic (\ooalign{\hfill{Y}\hfill\crcr{=}}{}500) - available from teacher,{\EdbZSp} 2nd week}%
\EdbColDef{edb:lecture.contact}{1}{E}{\�A����l{}{NIMCHUK . }{c{\EdbZSp}a{\EdbZSp}n{\EdbZSp}a{\EdbZSp}d{\EdbZSp}a{\EdbZSp}c{\EdbZSp}l{\EdbZSp}u{\EdbZSp}b{\EdbZSp}j{\EdbZSp}p{\EdbZSp}@{\EdbZSp}b{\EdbZSp}l{\EdbZSp}o{\EdbZSp}o{\EdbZSp}m{\EdbZSp}.{\EdbZSp}o{\EdbZSp}c{\EdbZSp}n{\EdbZSp}.{\EdbZSp}n{\EdbZSp}e{\EdbZSp}.{\EdbZSp}j{\EdbZSp}p{\EdbZSp}}}%
\��������{�p��(1)}%
\������{��Չp��}%
\end{edb:lecture}%

\end{document}
